Author’s response to reviews

Title: Quality of Intrapartum and Newborn Care in Tigray, Northern Ethiopia

Authors:
Girmatsion Fisseha (girmaf4@yahoo.com)
Yemane Berhane (yemaneberhane@gmail.com)
Alemayehu Worku (alemayehuwy@yahoo.com)

Version: 4 Date: 08 Dec 2018

Author’s response to reviews:

Dear editor

We would like to thank you again for your comments and suggestions to our manuscript. It was very constructive. Here is our point-by-point response

Comment and Responses

- There continues to be high text overlap to previously published papers in the beginning of the Background. Please rewrite this section in your own words.

Accepted and modified accordingly

- Please have your manuscript further edited for English usage by a colleague who is fluent in English or by a professional editing service. Please see below for more information.

Accepted and modified accordingly. We used the word “labour” than “labor” in this manuscript b/c the word “labour” is mostly found and used in medical literature. The word “partograph” and “intrapartum” are also correctly written in this manuscript since it is highlight us as red. Overall, we have made a modification on the English usage in this manuscript.

- Thank you for reducing the overlap in the Methods. However, more detail is needed, not just a reference to the previous paper. Please summarize the procedure and add more information.
Thank you, well accepted comments!

We have revised the manuscript again and summarized the data collection procedure.

Sincerely!